Author’s response to reviews

Title: Prevalence and multivariable factors associated with inadvertent intraoperative hypothermia in video-assisted thoracoscopic surgery: a single center retrospective study

Authors:

Yinan Li (liyinan0272@bjmu.edu.cn)
Hansheng Liang (doctorliangsh@sina.com)
Yi Feng (doctor_yifeng@sina.com)

Version: 2 Date: 21 Jan 2020

Author’s response to reviews:

Dear Yong G. Peng, MD, PhD,
Editor-in-Chief
BMC Anesthesiology

January 21, 2020

Re: view letter of manuscript no. BANE-D-19-00804_R1

Dear Editor,

Thank you very much for your letter. It is our honor to be accepted by your journal.

We have made a supplementary explanation to the Authors’ Contributions section (Line 15-18, page 12), and have made no other modifications.

We hope that the revised manuscript will be sufficient to make our manuscript suitable for publication in the BMC Anesthesiology. Looking forward to hearing from you soon. Thank you very much.

Sincerely,

Yi Feng, MD
doctor_yifeng@sina.com

Editor Comments:
1. Authors' contributions
-- We have noted that author [HL] and [YF] are missing in the listed authors' contributions. The individual contributions of ALL authors to the manuscript should be specified in the Authors’ Contributions section. Guidance and criteria for authorship can be found here: http://www.biomedcentral.com/submissions/editorial-policies#authorship

We have supplemented the specific contributions of all authors to the manuscript in the Authors’ Contributions section. (Line 15-18, page 12)
2. Additional files
-- Please remove any files from your file inventory that you do not wish to be published alongside your
manuscript.

We have submitted only the documents for publication

3. Clean copy
-- At this stage, please upload your manuscript as a single, final, clean version that does not contain any
tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant
tables/figures/additional files should also be clean versions. Figures (and additional files) should
remain uploaded as separate files. Please ensure that all figures, tables and additionalsupplementary
files are cited within the text.

We have uploaded the clean manuscript, tables and figures respectively.